EO-1 Weekly Status Week of April 1 – April 7, 2010
Day of Year 091 - 097
Mission Day 3428 - 3434
Earth Observing One (EO-1) General.
Scheduled 170 science Data Collection Events (DCEs) past week
INSTRUMENTS

All instruments operated nominally this week 
· Performed instrument calibration

· Conducted ALI internal calibration Type I on April 5 at 01:09z

EO-1 Spacecraft Subsystems

Command and Data Handling (C&DH)

Continued to experience problems playing back engineering data from the Solid-State-Recorders.  Engineering data is being received during real-time contacts and for all science imaging events. 

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

An unsolicited proposal has been submitted to Michael Goodman at NASA Headquarters who is the Application Lead for Disasters in the Earth Science Division.  This proposal is directed at sustaining CEOS, GEOSS, and GEO tasks.

On April 1, Stu Frye met with the GIS coordinator for Jamaica, Nikolai Thomas, to discuss the Caribbean Disaster Pilot Project and its utility to Jamaica’s needs.  The Canadian Space Agency and the European Space Agency work for the Caribbean Disaster Pilot has been funded and is underway now.  This meeting with Nikolai Thomas was a key in shaping the CSA and ESA assignments for the Pilot work in Jamaica.
Microtel will be doing testing of Delay-Tolerant Networking (DTN) on EO-1 in support of JPL’s work on space qualifying DTN. 

On Thursday, April 1, there was an ESTO AIST 2008 SensorWeb 3G collaborators teleconference.  

Participants:  Dan Mandl, Pat Cappelaere, Vuong Ly, and Joe Young.

Notes from this teleconference are as follows:

1.  The team is to focus on getting ready to support the GSFC HyspIRI Science Symposium on May 4-5 where a presentation will be made on the capabilities of the Intelligent Payload Module (IPM) to perform on-board processing of science data and to direct broadcast downlink processed or raw science data.
2.  Pat Cappelaere and Vuong Ly are working to run the FLAASH atmospheric correction algorithm on the SpaceCube processor.

3.  Vuong is to check performance of the current IPM design to determine if more processing power is needed.  
4.  In support of this assignment, a meeting will be arranged with Jonathon Wilmot at GSFC, who is an OPERA processor expert, to discuss the HyspIRI IPM data ingest and processing requirements and to determine how to compare SpaceCube processor performance with OPERA processor performance.
On Tuesday, April 6, there was a Namibia Flood/Disease SensorWeb collaborators teleconference.  

Participants:  Dan Mandl, Stu Frye, Guido van Langenhove, Lenny Roytman, Felix Kogan, and Joe Young.

Notes from this teleconference are as follows: 

1.  Dan Mandl stated that he has the following four to-do items:
· Convert the Flood/Disease SensorWeb Pilot Project Proposal work effort into “bite size chunks” so that donors can more readily determine what activities they can afford to support.
· Send a request letter to the Department of Survey in Namibia to obtain the database of locations of all houses in Namibia.

· Call Andrea Reich in Namibia to locate the most appropriate Health Ministry person to serve as an active collaborator in the heath area.

· Send UN-SPIDER a short description of NASA’s organization and role in the Namibia Pilot Project.
2.  Dan is to update Work Package 3 of the proposal.
3.  Yudong Tian, at the Univ. of MD, is to transfer his flood information database to Fritz Policelli’s server at GSFC.
4.  Guido van Langenhove stated that he needs to start receiving RADARSAT-2 images and continue receiving EO-1 images.  Stu Frye has arranged for doing RADARSAT-2 tasking and has been delivering 1-2 EO-1 images per week since March 1.
Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week. 

EO-1 MISSION OPERATIONS CENTER

Real-Time

· Continued nominal spacecraft state of health

· Monitoring WARP single-bit event messages to ensure a nominal frequency of single bit errors

Mission Planning

· Continuing testing of ASIST SCP to replace legacy CMS mission planning software

· Confirmed xml output contains information needed to verify commands

Flight Dynamics
· Completed troubleshooting of anomaly in PF1 tracking data, starting January 19

· Received notification from station stating that a bias was added inside PF1 to compensate for the tracking data errors instituted on April 6 at approximately 14:00z

· Analyzed subsequent data and determined that the bias has returned PF1 to pre anomaly values in tracking data

· Continuing to test file deliveries to/from new flight dynamics computer

· Resolving a problem in which too many emails were sent to a single facility due to multiple users receiving the files (limit 10/hour) 

· Working with USGS to institute a listserv on their end that we send messages to so they can add / remove emails at their leisure and EO-1 only sends up to 4 messages to USGS

· New flight dynamics computer continues to resend the unsent emails every 15 minutes until they are all accepted by the facility (all eventually sent)

· Worked with SGS to reestablish tracking data files from passes they conducted

· When SGS came back online from their scheduled downtime they only sent tracking data files to the FD group.

· SGS added back into automation a file transfer of tracking data files to EO-1 on April 2

· Confirmed that EO-1 is receiving the tracking data files with correct permissions

· Tested the tracking data files that Wallops delivered to the new flight dynamics computer

· Successfully processed the tracking data files from Wallops
Trending
No change from last report.
System Administration
· Resolved issue that Wallops was having with delivering tracking data to EO-1’s new Flight Dynamics computer. The Windows XP firewall was updated to with the Closed IONet to Open IONet proxy server’s IP address instead of the individual IP addresses of the computers at Wallops. The connection was tested successfully by Wallops personnel.

· Completed the following work on the Windows XP flight software development system on the private network.

· Installed the latest XP patches on the system.

· Applied CIS Benchmark settings to the computer.

· Installed the following updates to the CNE computers:

· Firefox 3.6.3

· Java 6 update 19

· Cumulative IE7 patch

· Windows XP patches

· Addressed the medium vulnerabilities found on EO-1’s four CNE printers by changing the SNMP community name so that it is no longer the default name provided by the vendor.

GROUND AND SPACE NETWORK

Station Downtimes

None
Operational Discrepancies
· Received notification that PF1 went green on April 6 at 14:00z from the tracking data anomaly starting January 19

· Starting receiving tracking data files from SGS on April 2

UPCOMING EVENTS

· Perform a pair of 350s inclination maneuver on April 13 at 12:52:40z and April 15 at 12:30:32z to maintain the MLT at very close to 10:00am

Imagery Status

Scenes and Engineering Cals planned for week of April 1 – April 7, 2010      170
Total scenes and engineering calibrations planned for entire mission – approximately 52,564
Total Scenes:  ALI scenes in the Level 0 archive              48,070                (as of April 7, 2010)

                         Hyperion scenes in the Level 0 archive    47,822                
Publications and Presentations Status 

404 publications 

284 external presentations

53 articles and press releases
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